# Threshold Effects in the Relationship between Inflation and Economic Growth: Evidence from Rwanda

By

Musoni J. Rutayisire IPAR-Rwanda

AERC Research Paper 293 African Economic Research Consortium, Nairobi February 2015

### 1. Introduction

#### Background

It is now widely accepted among economists, policy makers and central bankers that the main objective of macroeconomic policy is to achieve a high and sustained economic growth rate while maintaining a low inflation rate. It is also generally believed that high inflation is detrimental to medium and long-run economic growth. Not surprisingly, the existence and the nature of the relationship between inflation and economic growth have become the subject of an extensive body of theoretical and empirical studies (Temple, 2000). The debate revolves around the following questions: first, is the relationship between inflation and growth positive, negative or non-significant; second, how low should the inflation rate be not to have an adverse effect on economic growth?

Recent studies have uncovered that the relationship between inflation and growth may be non-linear. There was shown to be a positive relationship between low inflation and high output growth, while higher inflation was associated with lower economic growth. According to these studies, the hypothesis of non-linearity suggested that the adverse effect of inflation on economic growth is not universal; it appears only when inflation exceeds some turning point or threshold level below which inflation has a positive or non-significant impact on economic growth.

#### **Research problem and motivation**

Since the early 1990s, Rwanda has been implementing comprehensive economic stabilization and structural adjustment programmes with the financial and technical assistance of the International Monetary Fund (IMF) and other development partners. As is well known, in such IMF-supported programmes low inflation is a key element of the policy package, with the objective to achieve macroeconomic stability and a sustained high economic growth rate.

During the implementation of successive economic programmes, an inflation rate of around 5% was used as a policy target. This target was met most of the time, but was sometimes missed, notably due to external and internal supply shocks. Bearing in mind the growth-harming effects of very high as well as very low inflation rates (Ghosh and Phillips, 1998), the following question may be posed in the context of the Rwandan economy: was the inflation target chosen for policy purposes appropriate? In other words, was it higher or lower than an optimal or threshold level consistent with economic growth? Results from some empirical studies like Khan and Senhadji (2001) have suggested that the inflation threshold range is 1–3% for industrial countries and 11–12% for developing countries, while in a similar study by Kremer, Bick and Nautz. (2009) it was shown that the estimated inflation threshold is about 2.5% for industrial countries and 17% for developing countries. Moreover, these studies support the idea that when the existence of the threshold, and consequently the non-linearity in the relationship between inflation and growth, is neglected, substantial biases could affect the estimation of that relationship. Other estimates for specific developing and emerging countries suggested a wide range of inflation threshold levels: 10% for Lesotho (Seleteng, 2005); 9% for Pakistan (Mubarik, 2005); 6% for Bangladesh (Ahmed and Mortaza, 2005); 15% for Egypt (Kheir El-Din and Abou-Ali, 2008); 9% for Mexico (Risso and Carrera, 2009); 22.2% for Ghana (Quartey, 2010); 6% for India (Singh, 2010); 8% for Nigeria (Salami and Kelikume, 2010); and 4% for South Africa (Leshoro, 2012).

Moreover, the management of the National Bank of Rwanda has been envisaging a shift from a monetary-targeting to an inflation-targeting regime for conducting monetary policy. If the latter regime is adopted, a single level or a range of inflation target values must be determined. In this context, the present research may be helpful in the determination of such targets. The last impetus for this research was the availability of data on Rwanda spanning the period 1968–2010, which is more than four decades. This allows for a meaningful time series investigation and valid statistical inference.

Most studies that have been conducted on the inflation threshold effects on economic growth employed cross-sectional and panel data covering large samples from countries. These studies were justified by their ability to generalize empirical findings and their policy implications appeal. However, as suggested by Lin and Ye (2009) and Espinosa ,Leon and Prasad, (2010), due to the heterogeneous factors prevalent in different countries, it is important to carry out country-specific studies in order to relate empirical findings to policy designs in specific cases. Inflation threshold effects should then be estimated for each country separately, allowing the incorporation of country specific characteristics. In particular, this would provide useful information about the appropriate location and width of an inflation-targeting band. In line with these suggestions, this research is a contribution to the literature of developing country case studies on the existence of threshold effects in the relationship between inflation and economic growth.

As far as could be ascertained, no such research has been conducted to date using time series data for Rwanda. However, in two empirical studies<sup>1</sup> based on a panel data set comprising industrial and developing countries, including Rwanda, Kremer, Bick and Nautz (2009) showed that inflation is detrimental to economic growth in developing countries if it exceeds 17%, and that below that threshold the impact of inflation on growth is insignificant, while Pollin and Zhu (2005) estimated a threshold inflation range of 15–23% for low-income countries.

#### **Objective of the study**

The purpose of the study is to investigate the relationship between inflation and growth in the case of Rwanda and to determine whether there is a turning point or a threshold level of inflation, above which the inflation effect on economic growth

switches from positive or insignificant to negative. The investigation has been carried out by means of a quadratic regression model which is estimated as a second-degree polynomial. This widely used technique for estimating non-linear relationships allowed the identification of the turning point in the inflation-growth nexus. This approach has been used by Devarajan, Swaroop and Zou (1996), Hermes and Lensink (2001), Pattillo, Poirson and Ricci (2002), Clements, Bhattacharya and Nguyen. (2005), and Pollin and Zhu (2005). Annual data sets spanning the period 1968–2010 have been used for the empirical analysis in this research.

The remainder of this paper is organized as follows: section 2 presents a brief review of both theoretical and empirical literature on the relationship between inflation and growth; section 3 provides an overview of inflation and growth trends in Rwanda during the period covered by the study; section 4 discusses the methodological framework of this research , while section 5 describes the data used; section 6 presents the estimation results , while section 7 concludes and summarizes the policy implications of the study.

### 2. Literature review

n extensive body of theoretical and empirical studies investigated the relationship between inflation and growth both in the context of developed and developing countries. This section presents a brief review of the literature.

#### **Theoretical literature**

In the Keynesian model framework, comprising Aggregate Demand (AD) and Aggregate Supply (AS) curves, the AS curve is upward-sloping rather than vertical in the shortrun. The implication is that changes on the demand side of the economy resulting from expectations, labour force and policy actions such as discretionary monetary or fiscal policies, affect both prices and output in the short run, as predicted by the Phillips Curve (Blanchard and Kitoyaki,1987; Dornbusch, Fischer and Kearney, 1996; Romer, 2001). Therefore, the Keynesian model advocates the existence of a positive relationship between inflation and output. However, in this Keynesian framework it is not the case that inflation is itself a growth-enhancing force, rather, the point is that if rising aggregate demand leads to increased growth, then some inflation pressures are likely to emerge as a relatively benign by-product. The positive relationship between inflation and growth evident in short-run dynamics is unsustainable in the longer term and turns negative with a higher inflation rate.

Mundell (1965) and Tobin (1965) predict a positive relationship between inflation and capital accumulation, which in turn implies a positive impact on growth. The so-called Mundell-Tobin effect states that since money and capital are substitutable, an increase in the inflation rate erodes the purchasing power of money balances, which causes substitution between resources and leads to a shift in the portfolio allocation away from money balances to real assets. This will raise capital accumulation and thereby stimulate the economic growth rate (De Gregorio, 1996; Choi, Smith and Boyd, 1996).

It has been asserted that inflation may be positively correlated with growth since it can be considered a necessary grease or lubricant for the wheels of the economy. From this perspective, it is argued that under wage and price rigidities, a certain level of inflation can help to realign the relative prices in response to structural changes in production during fast modernization periods of the economy (Lucas, 1973; Akerlof, Dickens and Parry, 1996; Kiley, 2000). In this scenario, inflation is rather important for economic growth.

Focusing on the economies of developing countries, some economists pointed out that inflation contributes positively to economic growth as it induces savings and investment through a number of channels (Baer, 1967; Georgescu-Roegen, 1970; Taylor, 1983). Governments of developing countries, faced with inadequate public revenues, often resort to borrowing from central banks to finance their budget deficits. This seigniorage, or inflation tax resources, may be used by governments to increase capital formation by financing real investment. As long as this financing mechanism does not crowd out private-sector investment, inflationary finance would contribute to economic growth (the Kalecki effect).Nominal wages lag behind prices, due to slowly adjusting expectations,sluggish wage bargaining or Government repression. Consequently, it follows that inflation may boost economic growth by shifting income distribution from individuals to higher saving capitalist firms and hence increasing savings , investment and growth (the Kaldor effect).

Literature on the adverse effects of inflation on economic growth in the long-run is based on the idea that high inflation increases economic inefficiencies and reduces growth through the decrease of the level of investment as well as the rate of productivity growth (Fischer, 1993).

High and volatile inflation interferes with the price signalling mechanism, resulting in confusing information for economic agents on relative prices, which in turn induces distortions in investment decisions and hence impedes the efficient allocation of resources (Fischer, 1993; Huybens and Smith, 1998; Khan and Sendhadji, 2000, 2001). Inflation creates uncertainty in financial markets and increases the risk associated with investment. Financial intermediaries are not eager to provide long-term financing for capital formation and tend to maintain liquid portfolios, which translates into a reduction of economic activity (Boyd, Levine and Smith, 1996; Hellerstein, 1997; Romer, 2001). High inflation also causes "shoe leather costs", which are associated with additional efforts that people make to reduce their holding of cash, and "menu costs" that arise from the necessity to change prices more often.

Inflation inhibits financial development; an inflationary environment is often associated with financial repression, as governments take actions, such as implementing interest rate ceilings and credit allocation to protect some priority sectors of the economy. Such controls lead to inefficient resource allocation and hamper economic growth (Boyd, Levine and Smith, 1996; Haslag and Koo, 1999; Rousseau and Wachtel, 2002).

In endogenous growth theory, the economic growth rate depends on the rate of return on capital. Since inflation acts as a tax, it decreases the real rate of return, and it follows that inflation impedes capital accumulation and hence decreases the growth rate (Fama and Schwert, 1977; Boyd, Levine and Smith, 1996).

Inflation causes a real appreciation of the domestic currency and reduces international competitiveness by making exports more expensive. In a country with a fixed exchange rate, inflation could lead to the deterioration of the trade balance and capital outflows and impact negatively on the long-term economic growth (Dollar, 1992; Easterly, 1999). Moreover, inflation can interact with the tax system to distort borrowing and lending decisions, thereby raising the cost of capital and reducing the real rate of return that discourages investment, ultimately reducing economic growth (Feldstein, 1982; Jones and Manuelli, 1993).

#### **Empirical evidence**

A look at the empirical evidence on the inflation-growth nexus reveals that results vary across time depending on data periods, country experiences and research methodology.

Earlier works (Bhatia, 1960; Dorrance, 1966; Johanson, 1967) found the relationship between inflation and growth to be either non-significant or positive. Based on data from the 1950s and the 1960s, these findings were in line with the prevailing view that inflation was not an important issue. As pointed out by many authors (Sarel, 1996; Cuaresma and Silgoner, 2003), during this period high inflation was basically unknown in most countries and the growth costs of inflation were not considered a serious problem. This is why empirical studies failed to establish any meaningful relationship between inflation and economic growth.

However, after the two oil price shocks (in 1973 and 1979) and the emergence of severe periods of high and persistent inflation rates in many countries during the 1970s and 1980s, the traditional point of view changed radically. As more data became available for these periods, several empirical studies repeatedly confirmed that inflation has a significant and negative effect on economic growth.

In a cross-country study conducted by Kormendi and Meguire (1985) using data of 47 sample countries for the period 1950–1977, it was found that an increase of inflation by 1% reduces economic growth by 0.57%. Fischer (1993) showed that there is a negative relationship between economic growth and some macroeconomic indicators, notably inflation and budget deficits. He further pointed out that the causality runs from these macroeconomic indicators to economic growth.

Barro (1995) explored the inflation-growth nexus using panel data for 100 countries over the period 1960–1990. His empirical finding was that there exists a statistically significant negative relationship between the two variables. He estimated that an average increase in inflation of 10% reduces output growth by 0.2% to 0.3%. In a cross-country study with a data set covering the same period, Motely (1998) detected a similar relationship and his finding was that an increase in inflation of 5% results in a decrease of economic growth of 0.1% to 0.5%.

Regional empirical studies confirmed the existence of a negative relationship between inflation and economic growth: De Gregorio (1992) for Latin America; Fischer, Sahay and Vegh (1996) for transition economies; Gillman, Harris and Matyas (2004) for OECD and APEC countries. The main finding of these studies was that inflation impedes efficient resource allocation by distorting the signalling role of price changes and by producing a variety of output-reducing inefficiencies.

It emerged from the above studies that the effect of inflation on economic growth is positive or non-significant at low rates, but this effect becomes significantly negative at higher rates. It follows from these findings that policy makers should aim for low rates of inflation to foster economic growth. But how low should the inflation be? In other words, at what level does inflation become detrimental to output growth? The answer to the latter question obviously depends on the structure and level of development of the economy and will vary from one country to another. Several empirical studies conducted since the mid-1990s have examined this issue, focussing specifically on whether the relationship between inflation and economic growth is non-linear. It was hypothesized that if such a relationship exists, it should be possible to estimate the threshold or the structural breakpoint at which the sign of the relationship between the two variables switches from positive to negative.

Fischer (1993) was the first to investigate the possibility of non-linearities in the relationship between inflation and output growth using both cross-sectional and panel data of 93 countries, including developing and industrial countries. He found a positive relationship between inflation and economic growth at low inflation rates, but the relationship became negative as inflation rose. Moreover, using two structural breakpoints, 15% and 40%, Fischer showed not only the presence of non-linearity in the relationship between the two variables, but also that the strength of the relationship weakens for inflation rates above 40%.

Following the results of Fischer (1993) there has been an expanding body of empirical evidence that shows the relationship between inflation and long-run growth was characterized by non-linearities and the existence of threshold effects.

Sarel (1996) used panel data from 87 countries covering the period 1970–1990 and tested for the existence of a threshold effect between inflation and growth. He found evidence of a structural breakpoint at an annual inflation rate of 8%. Below that rate, inflation does not have a significant effect on economic growth, or it may even show a marginally positive impact. Above that level, the effect is negative, statistically significant and very strong. Ignoring the existence of the threshold would substantially bias the impact of inflation on growth.

Ghosh and Phillips (1998) found that although inflation and growth are positively related at very low inflation rates (about 2% to 3% a year), the relationship is reversed at higher rates. Furthermore, the relationship is convex, so that a decline in the growth rate associated with an increase in inflation from 10% to 20% is greater than the fall in growth following a move in inflation from 40% to 50%. This finding confirmed Fischer's (1993) results. They also found that inflation is one of the most important statistical determinants of growth.

Bruno and Easterly (1998) examined the determinants of economic growth using cross-sectional data from 26 countries for the period 1961-1992. They argued that the negative relationship between inflation and growth exists only in high frequency data and with extreme inflation observations. In their empirical analysis, they detected a threshold level of 40%, above which the relationship between inflation and growth was negative. In addition, they found an inconclusive relationship between inflation and economic growth below this threshold level when countries with high inflation crises were excluded from the sample.

Khan and Senhadji (2000, 2001) investigated the inflation-growth interaction for both developing and industrial countries separately, applying the threshold panel data estimation technique originally developed by Hansen (1996,2000). They used a panel data set from 140 countries covering the period 1960–1998. Their findings strongly suggested the existence of a threshold level beyond which inflation exerts a negative effect on output growth. The threshold level was 1-3% for industrial countries and 11-12% for developing countries, respectively. The negative and significant relationship between inflation and growth above the threshold level was quite robust with respect to the estimation method and different specifications. The results clearly suggested that the threshold level is lower for industrialized countries than for developing countries.

Sepehri and Moshiri (2004) also tested also the non-linearities in the inflation-growth nexus for industrial and developing countries. Using a non-linear specification and data from four groups of countries at various stages of development, they found that the turning points varied widely, from as high as 15% for lower-middle-income countries to 11% for low-income countries, and 5% for upper-middle-income countries. No statistically significant long-run relationship between inflation and growth was detected for OECD countries. Their findings also pointed out the potential bias in the estimation of the inflation-growth nexus that may result from combining various countries, Pollin and Zhu (2005) found threshold inflation ranges of 14–16% for middle-income countries and 15–23% for low-income countries.

Working with data of a panel from 124 industrial and developing countries, Kremer, Bick and Nautz (2009) investigated the presence of threshold effects of inflation on long-term economic growth. Their empirical results showed that the estimated inflation threshold level was about 2.5% for industrial countries and 17% for developing countries. Above these critical levels, the inflation rate leads to a lower long-term economic growth rate in both cases. In addition, the study indicated that below these thresholds, the effect of inflation on long-term economic growth was significantly positive in developed countries. By contrast, there was no significant impact on economic growth in developing countries when inflation was below 17%.

In a recent study using panel data for the period 1980–2008, Seleteng, Bittencourt and van Eyden (2011) examined the growth-inflation nexus for the SADC<sup>2</sup> countries and provided new evidence on the existence of threshold effects of inflation on growth in African economies. By applying a panel smooth transition regression model, the findings of the study revealed a threshold level of 18.9% above which inflation is detrimental to economic growth in the SADC region. The model also estimated the smoothness of the transition from a low to a high inflation regime.

It is evident from the above empirical literature review that most studies on the threshold effects of inflation on economic growth used cross-sectional or panel data covering a large number of countries. However, a few studies on emerging and developing countries applying the methodology developed by Khan and Senhadji (2000, 2001) used time series data to estimate the threshold level of inflation for country-specific cases. These include, among others: Mubarik (2005) for Pakistan; Seleteng (2005) for Lesotho; Ahmed and Mortaza (2005) for Bangladesh; Kheir El-Din and Abou-Ali (2008) for Egypt; Risso and Carrera for Mexico (2009); Salami and Kelikume (2010) for Nigeria; Quartey (2010) for Ghana; Singh for India (2010); and Leshoro (2012) for South Africa. These studies confirmed the existence of an inflation threshold effect on economic growth in the different country-specific cases.

## 3. Inflation<sup>3</sup> and growth trends in Rwanda

During the period covered by this research, the Rwandan economy experienced a mixed performance: economic growth and inflation have been characterized by notable fluctuations, resulting from the combined effects of domestic and external factors. The domestic factors included, in particular, supply shocks due to recurrent climate conditions, the breakdown of Rwanda's system of production and distribution during the war in 1990–1994, and the demand shocks reflecting the effects of monetary and fiscal policies. Regarding external factors, the period under consideration has been marked by two oil price shocks, in 1973–1974 and 1979–1980, and the more recent combined increase in prices of oil and food products in 2003–2008.

As a first step in exploring the bivariate relationship between growth and inflation, Figure 1 illustrates the historical trends of the two macroeconomic variables. The annual average real economic growth rate for the whole sample period was 5.4%, while the average inflation rate was 9.5%. As can be seen from the graph in Figure 1, the Rwandan economy witnessed two severe recessions and the highest inflation rates in 1974 and 1994 reflecting, respectively, the oil price shock in 1973–1974 and the collapse of the national economy at the end of the war in 1994. As a consequence of these shocks, the real GDP contracted by 11.3% in 1974 and 64.5% in 1994, while inflation was as high as 32.6% and 64% in the same periods. In both cases, the slowdown of overall economic activity was followed by a strong and sustained recovery in subsequent years. However, the recovery was accompanied by higher inflation rates reflecting expansionary macroeconomic policies implemented by the government to stimulate economic growth.

Comparing the economic growth performance during the two sub-periods 1968–1989 and 1990–2010, it appears that the first sub-period witnessed a lower average growth rate of 4.5% against 6.4% observed in the second sub-period. The performance would have been even better in the sub-period 1990–2010 if a severe economic recession due to the genocide in 1994 was not accounted for. The significant difference in economic growth and the behaviour of other economic indicators observed in the two sub-periods can be explained to a large extent by the drastic changes in economic management that occurred in Rwanda between the 1990s and 2000s.

During the 1970s and 1980s, the Rwandan economy was characterized by heavy government intervention and regulation. Not only did the government own and manage an important economic portfolio, but it also determined the prices of goods and services. The financial system functioned according to the McKinnon-Shaw repression paradigm, which is characterized by government interference in the operations of the financial system through interest rate ceilings and direct credit control. In this context, the negative effects of inefficient resource allocation were aggravated by the shocks from the increase

in oil prices and the fall in prices of Rwanda's main export products (coffee and tea), and resulted in lower growth and unsustainable macroeconomic imbalances in the late 1980s.



Figure 1: Inflation trends and real GDP growth

The period between 1990 and 2010 was deeply marked by the genocide and the collapse of the Rwandan economy. However, it also witnessed important economic reforms, which allowed Rwanda not only to restore financial stability and recover economic activity, but also to gradually transition from a state-controlled to a market-oriented economy relying on the private sector. These reforms were implemented through successive economic stabilization and adjustment programmes – Structural Adjustment Program (1990), Enhanced Structural Adjustment Facility (1998), Poverty Reduction and Growth Facility (2000) and Policy Support Instrument (2010) – which focused on domestic revenue mobilization, strengthening public finance management, enhancing the effectiveness of monetary policy, relaxing exchange restrictions and removing impediments to private-sector development. Along with these reforms, priority spending aimed at promoting high economic growth and reducing poverty were channelled to investment in economic sectors such as infrastructure and agriculture, as well as to social sectors like health and education. It is worth mentioning that these reforms were supported by large inflows of foreign aid and international technical assistance.

As shown in Table 1, investment and government expenditure as a share of GDP recorded a steady increase, interrupted by the period of war, while the removal of exchange rate restrictions improved the openness of the economy and boosted the external sector. As can be seen in Table 1, the implementation of sound economic policies resulted in all economic indicators performing better, on average, in the second sub-period 1990–2010 than in the first sub-period 1968–1989, even though higher inflation was experienced in the second sub-period reflecting expansionary macroeconomic policies implemented by the government to stimulate economic growth and reduce poverty.

<sup>&</sup>quot;INF" = inflation; "Growth" = economic growth.

Indicators	1968–1979	1980–1989	1990–1999	2000–2010
Real GDP growth rate	6.4	2.5	4.6	8.1
Inflation rate	10.3	4.7	15.4	7.6
Investment/GDP	11	15.3	14.5	20
M3/GDP	13.5	14.5	15.9	18.1
Government⁴	12.4	13.2	11.3	17.5
Openness	29.8	31.2	32.1	36.5

Table 1:	Economic indicator	trends	3
----------	--------------------	--------	---

Source: Original data on different variables are from World Bank Development Indicators (2011) and National Bank of Rwanda (various issues of Annual Reports and Monetary Surveys); ratios and rates are author's calculations.

Regarding the nature of the relationship between inflation and growth, no obvious conclusion can be drawn on the basis of a visual inspection of the historical trends between the two variables depicted in Figure 1. For example, there is no evidence that higher inflation correlates systematically with lower economic growth or the reverse. It emerges instead that the correlation looks either positive or negative in the sense that the two variables are moving in the same or in the opposite direction, depending on the sub-periods of the sample.

This ambiguous relationship between inflation and growth is confirmed by the data presented in Table 2. Following the approach used by Mubarik (2005) and Ghosh and Phillips (1998), sub-ranges of inflation rates have been computed and categorized in ascending order. Average inflation and growth rates corresponding to each inflation range have also been estimated.

No clear pattern emerges from the distribution of the average inflation and growth rates. It is worth recalling that, on average for the whole sample period, as indicated in Table 1, the inflation rate varied between 4.7% and 15.4%, while the economic growth rate ranged between 2.5% and 8.1%.

Inflation ranges	Number of observations	Average inflation	Average growth rate
1) -2.5 <π,< 2	9	0.23	6.3
2) 2 < $\pi_{t}$ < 4	7	3.52	2.7
3) 4 < π < 6	4	6.11	4.9
4) 6 < π, < 8	4	7.35	11.0
5) 8 < π <sub>.</sub> < 10	6	9.28	7.1
6) 10 < π, < 12	2	11.86	12.0
7) 12 < π, < 14	4	12.96	-2.5
8) 14 < π <sub>t</sub> < ∞	7	28.14	5.9

Table 2:	Inflation	ranges	and	economic	growth
----------	-----------	--------	-----	----------	--------

Source: Author's calculations

It can be observed that from the first to the third inflation ranges, a higher average inflation rate leads to lower economic growth, suggesting a negative relationship. However, the higher average inflation rates corresponding to the fourth, fifth and sixth inflation ranges coexist with impressive average economic growth rates varying between 7% and 12%. These observations provide some preliminary evidence that there may be a non-linear relationship between growth and inflation, and raises the question regarding the sign of the relationship between the two variables and the level of the inflation threshold.

### 4. Methodological framework

This section discusses the methodological framework adopted for the empirical analysis of this research, specifies the model, and highlights the econometric approach.

#### General growth model

The starting point of the investigation into the threshold effects on the inflationgrowth nexus in Rwanda is a general model describing the link between economic growth and its determinants. This model takes the form of a growth regression equation augmented with inflation:

$$\Delta \ln y_t = \beta_0 + \beta_1 \pi_t + \beta_2 X_t + \varepsilon_t \tag{1}$$

where  $\Delta lny_t$  is the annual real GDP growth, computed as the log first difference of the annual real GDP and  $\Delta$  is the first difference operator;  $\pi_t$  represents inflation;  $X_t$  is the vector of explanatory variables, selected among the most commonly used in the growth literature;  $\beta_2$  is the matrix of parameters of the explanatory variables;  $\beta_0$  is a constant; and  $\epsilon_t$  is the error term.

A common problem in growth theory is the determination of the main sources of growth or the choice of the set of explanatory variables to be included in Equation 1. Neoclassical growth theory focuses on capital stock, labour force and technological progress as the main driving forces of growth (Solow, 1956; Swan, 1956), while in the endogenous growth theory, a crucial role is given to human capital, knowledge and new technologies (Romer, 1990; Grossman and Helpman, 1991). This research draws upon the neoclassical model to justify the choice of basic explanatory variables. The power of the Solow model to explain growth in both developed and developing countries has been widely documented in the literature dealing with the empirics of economic growth (Mankiw, Romer and Weil ,1992). However, due to data constraints, this analysis will substitute investment and population for capital stock and labour force, respectively, in the empirical model.

In the empirical literature, Levine and Renelt (1992) and Sala-i-Martin (1997) argue that despite the existence of a large set of explanatory regressors that can potentially be used in the growth regression, only a few of them may be significant. They further proposed checking the robust regressors econometrically.

As a result of Sala-i-Martin's test for robustness, the following explanatory regressors have been identified as among the most important determinants of growth: investment, population growth, inflation rate, government expenditure, trade openness and the growth rate of terms of trade. These variables have in common that they are systematically correlated with growth. Financial development is another important variable that has been emphasized by empirical studies in the growth process (King and Levine, 1993; Levine and Zervos, 1998), hence this variable has also been considered in this study. Therefore, apart from inflation, the empirical analysis of this research into Rwanda will rely on the above results and will use the following basic model:

$$\Delta \ln y_t = \beta_0 + \beta_1 \pi_t + \beta_2 \ln V_t + \beta_3 \Delta \ln POP_t + \beta_4 OPEN_t + \beta_5 FD_t + \varepsilon_t$$
(2)

where  $\Delta \ln y$  is as defined in Equation 1;  $\pi$ , representing inflation is defined as the annual percentage change of the Consumer Price Index (CPI,); INV, stands for investment proxied by gross fixed capital formation as a share of nominal GDP;  $\Delta$ InPOP, is the annual population growth rate calculated as the first log difference of population; OPEN, represents trade openness calculated as the ratio of the sum of exports and imports to nominal GDP, and FD, is a measure of the financial development indicator computed as the ratio of liquid liabilities of the financial system (broad money represented by the M3 monetary aggregate) to nominal GDP, and  $\beta_0$  and  $\epsilon_t$  have the same definition as in Equation 1. Two dummy variables (Dummy1 and Dummy2) and a trend term have been added to the list of regressors. The dummy variable, Dummy1, takes on the value 1 in 1992–2005 and zero otherwise; it is designed to account for the influence on growth of the comprehensive economic reforms implemented in Rwanda during the 1990s and 2000s. The dummy variable, Dummy2, is set to 1 in 1991–1994 and zero otherwise; it is meant to capture the destabilizing effect of the war period and genocide that led to the collapse of the Rwandan economy. A trend term has also been added to the model to reflect the impact of other shocks which may have affected economic growth in the sample period.

The set of growth determinants adopted in this research has also been used in some influential empirical works investigating the relationship between inflation and growth (Sarel, 1996; Khan and Senhadji, 2001; Mubarik, 2005 and Kremer, Bick and Nautz, 2009) as well as in recent works on developing economies (Risso and Carrera, 2009; Frimpong and Oten-Abayie, 2010; Seleteng, Bittencourt and van Eyden, 2011; Ghazouani, 2012; Ayisi, 2013; Ahortor, Adenekan and Ohemeng, 2012).

Theoretically, the expected effects of the different explanatory variables on growth in Equation 2 are as follows: Investment has been identified by both neoclassical and endogenous growth models as the most fundamental determinant of economic growth, hence the marginal impact of investment or physical capital ( $INV_t$ ) is expected to be positive, since a change in the investment-to-GDP ratio (I/Y) will lead to an increase in growth according to the accelerator principle. The Harrod-Domar model also argues that the growth rate of output depends on the level of savings and the share of investment spending in GDP.

The growth rate of the population ( $\Delta$ lnPOP<sub>1</sub>) may have either a positive or a negative effect on output growth. According to Todaro (1996), larger populations provide the required aggregate demand ("size" effect) to generate favourable economies of scale in production, lower production costs and provide sufficient labour to achieve higher output levels. Moreover, positive effects of population growth also stem from the human capital contribution. However, as Kelley (1988) pointed out, population growth could have a negative impact on economic growth if the dependency of the young population lowers investment ("age dependency" effect) as a result of diverting resources from more productive activities to basic needs, or if the average productivity of physical capital is lowered via diminishing returns ("capital-shallowing" effect).

It is now widely agreed that financial sector development (FD<sub>t</sub>) contributes to economic development as it promotes economic growth through the following functions performed by the financial system: mobilization of savings, efficient allocation of resources, management and pooling of risks, liquidity provision, and reduction of information and transaction costs (Levine, 1997). King and Levine (1993) provide empirical evidence that higher financial development is associated with faster growth. Levine, Loayza and Beck (2000), and Rousseau and Wachtel (2002) came to the same conclusion: that financial development has a positive impact on growth. In growth literature, the ratio of broad money to GDP, representing the size of financial depth, is one of the most used measures of financial development (Hussain, 2005; Frimpong and Oteng-Abayie, 2010).

Regarding openness (OPEN<sub>t</sub>), theory and empirical studies have shown that trade does not have a simple and straightforward relationship with economic growth. Some economists have contended that trade can act as "an engine" of growth and highlighted the beneficial effects that promoting exports has on overall economic activity (Sachs and Warner, 1995; Edwards, 1998). By contrast, others have emphasized the monetary and balance-of-payments consequences of trade. They pointed out the loss of output that would result from a restrictive adjustment policy to adjust the disequilibrium of balance of payments that is not self-correcting through relative prices movements, i.e., the real exchange rate (Thirlwall, 2000). It follows from these contrasting views that the final effect of openness on growth is an empirical issue.

#### Modelling inflation threshold effects on growth

To investigate the existence of a non-linear relationship between inflation and economic growth, most empirical studies use the threshold endogenous model developed by Sarel (1996) and Khan and Senhadji (2001). However, this model requires a large set of data to make valid statistical inferences. Therefore, given the relatively small size of the sample in the present study (43 observations), and following Pollin and Zhu (2005) and Quartey (2010), the quadratic function has been adopted to estimate the threshold level or the turning point above which inflation exerts a negative effect on economic growth in the case of Rwanda. The quadratic function approach has also been used in other areas of empirical growth literature. For example, Pattillo, Poirson and Ricci. (2002) and Clements, Bhattacharya and Nguyen . (2005) employed the quadratic function to estimate the non-linear relationship between external debt and growth, while Devarajan, Swaroop and Zou. (1996) and Hermes and Lensink (2001) relied on the

same approach to determine the optimal size of government, i.e., the share of overall government spending that maximizes economic growth. In line with these works, the following quadratic function has been used in the present study to examine the non-linear relationship between inflation and growth:

$$\Delta \ln y_t = \beta_0 + \beta_1 \pi_t + \beta_2 \pi_t^2 + \beta_3 \ln V_t + \beta_4 \Delta \ln POP_t + \beta_5 OPEN_t + \beta_6 FD_t + \varepsilon_t$$
(3)

in which the squared term of inflation,  $\pi_t^2$ , has been added, while all other variables are as defined in Equation 2. As this research focuses on the relationship between inflation and growth, investment as a share of GDP, population growth, openness, and financial development indicators are used as control variables.

In this model, it is expected that the linear term of inflation,  $\pi_t$ , would have a positive sign and is designed to reflect the beneficial effects of low inflation on output, while the squared term of inflation,  $\pi_t^2$ , is expected to have a negative sign and should measure the adverse impact associated with higher inflation. Since the squared term increases in value faster than the linear term, it implies that the presence of negative effects of inflation will eventually outweigh the positive effects. Moreover, the combination of a positive and significant linear term with a negative and significant squared term suggests that the impact of inflation on output can be described as an inverted U-shaped curve. This supports the view that the positive effects of inflation switches to negative when inflation exceeds a threshold level. The peak of the quadratic function identifies the inflation threshold level or the turning point above which the marginal effect of inflation becomes negative.

In order to find whether the hypothesis of non-linear effect of inflation on growth is confirmed, Equation 3 is estimated and the significance of the coefficients of the linear and squared terms is assessed. If both coefficients are significantly different from zero, we can find out the peak of the quadratic function that identifies the critical point of inflation above which the marginal impact of inflation on growth is negative. To calculate the critical point corresponding to the inflation threshold level, the partial derivative of Equation 4 is computed with respect to inflation,  $\pi_{t}$  and is set to zero:

$$\delta \Delta \ln y_t / \delta \pi_t = \beta_1 + 2 \beta_2 \pi_t = 0 \tag{4}$$

Solving the above equation for  $\pi_t$ , the critical point of inflation beyond which the marginal impact of inflation becomes negative is obtained:

$$\pi_t^* = -\beta_1 / 2 \beta_2 \tag{5}$$

## 5. Data, empirical results and discussion

This section presents the variables used in this research and their time series properties, explains the sources of the data and includes discussions on the estimation results of the study.

#### Data used

The basic data used in this study include time series on GDP, population, the CPI, gross fixed capital formation (as a proxy for investment or physical capital), imports and exports, broad money aggregate, and M3 (measure of liquid liabilities of the financial system). The set of data spanning 1968–2010 was collected from the World Bank Development Indicators and from various documents published by the National Bank of Rwanda and the National Institute of Statistics of Rwanda (Statistical Bulletins and Annual Reports). The nominal gross domestic product valued in domestic currency (Rwandan francs) has been deflated by the consumer price index to obtain the real GDP and the base year for the CPI is 1990 = 100.

#### Time series properties of variables

As the Ordinary Least Squares estimation technique has been adopted, it has to be ensured that all the variables included in the different models are stationary in order to have consistent results and to avoid spurious regressions. To this end, the time series properties of the variables have been investigated and the order of integration of each variable has been determined by applying the Augmented Dickey-Fuller (ADF) and Phillips-Perron (PP) unit root tests with "constant" and "constant and trend". These tests are complementary, as the PP generalizes the ADF test and provides robust estimates in the presence of serial correlation, time dependent heteroskedasticity and structural breaks in the time series. It is important to recall that some variables are defined as percentage annual changes or growth rates ( $\Delta lny_t$ ,  $\pi_t$  and  $\Delta lnPOP_t$ ), while others are defined as ratios (INV<sub>c</sub>, OPEN, and FD.). The results of both unit root tests are presented in Table 3.

According to the results in Table 3 of both unit root tests, ADF and PP, for all the variables used in this paper, the null hypothesis of non-stationarity has been rejected at a 5% significance level, meaning that these variables are stationary, i.e. I (0). Therefore, the relationship established between economic growth rate and the other variables would not be spurious.

Variables	ADI	F test	PP test		
	Constant	Constant and trend	Constant	Constant and trend	
∆lny,	-7.359**	-8.741**	-8.131**	-8.115**	
INV,	-10.40**	-10.83**	-3.778**	-6.296**	
π,	-4.888**	-6.092**	-4.029**	-3.959**	
π,2	-4.888**	-6.092**	-5.379**	-5.314**	
∆InPOP,	-5.064**	-4.750**	-3.511**	-3.698**	
OPEN,	-3.263**	-5.087**	-4.812**	-5.236**	
FD,	-2.997**	-4.842**	-3.188**	-4.572**	
Critical values:					
5% significance level	-2.94	-3.53	-2.935	-3.524	

Note: \*\* indicates that the null hypothesis of non-stationarity is rejected at 5% significance level.

#### **Causality test**

Before estimating the quadratic regression model and following a common practice in the literature exploring the relationship between inflation and growth, the Granger Causality test has been performed to investigate the linear causation between inflation and economic growth in Rwanda. This test is used because inflation may not be an exogenous variable in the growth-inflation regression and, consequently, the inflation coefficient may be biased. As pointed out by Khan and Senhadji (2001), the seriousness of this problem will depend, to a large extent, on whether the causality runs mainly from inflation to growth, in which case the endogeneity problem may be benign, or the other way around, in which case the problem may be more severe. However, as suggested by Fischer (1993), the causality is more likely to run predominantly from inflation to economic growth, implying that the bias will not be important and the inflation coefficient will not seriously be affected. This assumption has been explicitly tested through the Granger Causality test and the estimation results are presented in Table 4. It is important to point out that causality does not necessarily imply exogeneity. However, the absence of a feedback effect from growth to inflation helps in the choice of dependent and independent variable for the model specification (Mubarik, 2005; Risso and Carrera, 2009; Hussain and Malik, 2011). It also provides a good indication that inflation does indeed have an impact on growth (Chimobi, 2010).

Table 4:	Causality	tests
----------	-----------	-------

Pairwise Granger Causality Tests Sample: 1968–2010 Lags: 1				
Null Hypothesis:	Obs.	F-Statistic	Prob.	
Inflation does not Granger Cause Growth Growth does not Granger Cause Inflation	41 0.31409	3.16621 0.5785	0.0832	

The test statistic indicates that the null hypothesis that inflation does not cause economic growth is rejected at the 5-10% significance level, as shown by the p-value of 0.0832. This implies that inflation causes economic growth. Conversely, the null hypothesis that growth does not cause inflation is not rejected, meaning that economic growth does not cause inflation. It may then be concluded from these results that there is no feedback from output growth to inflation to suggest that there is a unidirectional causality running from inflation to growth.

#### Quadratic regression model results

 $\mathbf{F}$  ollowing the conclusions of extant literature, it was hypothesized that in Rwanda inflation exerts a negative effect on economic growth when it exceeds a certain threshold. To this end, a quadratic regression model was estimated to determine the level of the threshold.

Given the relatively small sample size (43 observations) and the use of annual data in this research, the number of lags was restricted to two in the estimation of the quadratic regression model in order to preserve the degrees of freedom. Hendry's (1995) dynamic general-to-specific approach was then adopted to derive a satisfactory model, which involves testing down the general model by successively eliminating statistically insignificant regressors to obtain the final parsimonious equation. In this process, population growth was not statistically significant, and therefore it was dropped.

The regression is estimated by means of an ordinary least squares technique. According to the results reported in Table 5, the goodness of fit of the model as reflected by the R<sup>2</sup> coefficient of determination is satisfactory. The quadratic regression model explains more than 97.2% of the variation in growth, and the t-values indicate that the main explanatory variables are statistically significant. The estimated Durbin Watson (DW) statistic rejects the hypothesis of serially correlated errors. Moreover, the F-statistic shows that the explanatory variables are jointly significant at the 5% significance level and the model successfully passes the conventional battery of diagnostic tests for serial correlation, normality, autoregressive conditional heteroskedasticity (ARCH), heteroskedasticity, and functional form, implying that the model is well specified and the results are valid for reliable interpretation.

In line with theoretical predictions and empirical evidence, both investment as a share of GDP (INV<sub>t</sub>) and the financial development indicator (FD<sub>t</sub>) are statistically significant in the model and have a positive impact on growth, while the openness variable (OPEN<sub>t</sub>) is inversely related to growth. A change of 1% in investment as a share of GDP leads to an increase in the real output growth of 1.59%, while a change of 1% in openness reduces the output growth by 0.47%. It is worth noting that the positive impact of financial development on output growth appears with a one-period lag in the model, while the contemporaneous effect is negative. However, the cumulative effect of this variable on growth is positive.

The two dummy variables, Dummy1 and Dummy2, that are added to the model are both significant and have expected signs .The effect of the first dummy, Dummy1, is positive, capturing the growth-enhancing impact of the economic reforms implemented in Rwanda during the 1990s and 2000s, while the effect of the second dummy, Dummy2, is negative, reflecting the destabilizing effect of the period of war and genocide in 1990–1994. Finally, the trend term, which attempts to account for other shocks that may have affected growth, is also significant.

The estimation sample is: 1971–2010								
	Coefficien	t	Std.Error	t-value	t-prob	Part.R^2		
Constant	-0.0940325	5	0.09816	-0.958	0.346	0.0307		
INV,	1.59015		0.3857	4.12	0.000	0.3696		
FD,	-0.906142		0.5258	-1.72	0.095	0.0929		
FD <sub>11</sub>	1.45356		0.6835	2.13	0.042	0.1349		
OPEN,	-0.468994		0.1575	-2.98	0.006	0.2341		
π,	0.447460		0.2395	1.87	0.072	0.1074		
$\pi_t^2$	-3.19526		0.5657	-5.65	0.000	0.5238		
$\pi^{2}_{t-1}$	1.43867		0.3303	4.36	0.000	0.3955		
Dummy1-2	0.0682334	ŀ	0.02262	3.02	0.005	0.2388		
Dummy2	-0.0937167	7	0.03853	-2.43	0.021	0.1694		
Trend	-0.0042894	9	0.002031	-2.11	0.043	0.1333		
R^2	0.971899							
F (10, 29)	100.3 [0.00	0]*	*					
log-likelihood	78.7526							
DW	2.04							
no. of observations	40							
no. of parameters	11							
Diagnostic tests								
AR 1-4 test:	F (4, 25)	=	1.6093 [0.2031]					
ARCH 1-1 test:	F (1, 27)	=	0.019707 [0.8894]					
Normality test:	Chi^2(2)	=	0.0094802 [0.9953]					
Hetero test:	F (17, 11)	=	0.39764 [0.9573]					
RESET test:	F(1,28)	=	1.9589 [0.1726]					

Table 5: Results o	f the	quadratic	regression	model
--------------------	-------	-----------	------------	-------

Modelling  $\Delta lny_t$  by OLS

Regarding inflation, the coefficient of the linear term,  $\pi_t$ , has a positive sign, while the coefficient of the squared term,  $\pi_t^2$ , has a negative sign, as expected in this model. Moreover, both coefficients are statistically significant. These results suggest that low levels of inflation are growth-enhancing, while higher levels reduce growth, which implies that in this research, the relationship between inflation and economic growth is non-linear and may be described by an inverted U-shaped curve. In other words, these findings confirm the hypothesis that there does exist a turning point or threshold level of inflation above which inflation is detrimental to economic growth in Rwanda.

On the basis of the estimation results presented in Table 5, the threshold level of inflation was obtained in two steps: first, the partial derivative of the quadratic regression model was computed with respect to inflation,  $\pi_t$ ; second, the partial derivative was set equal to zero and then solved for  $\pi_t$  to find the turning point,  $\pi_t^*$ . In solving for  $\pi_t$ , the steady state values were used, whereby current and lagged values

of inflation and squared inflation are taken into account. The calculation yielded the following results:

$$\delta \Delta \ln y_t / \delta \pi_t = 0.44746 - 2(3.19526 - 1.43867) \pi_t = 0 \tag{6}$$

$$\pi t^* = 0.44746 - 2(1.75659) \pi_t = 0 \tag{7}$$

It follows that the threshold level of inflation is:

$$\pi_t^* = 0.44746/3.51318 = 12.7\% \tag{8}$$

The above result suggests that economic growth peaks when inflation reaches the threshold level of 12.7%, corresponding to the growth maximizing inflation rate. It follows that in the context of Rwanda, inflation is growth-enhancing when it is below 12.7%, while its impact on economic growth becomes negative when it is higher than the threshold level.

This finding is consistent with the results of similar empirical studies on developing countries (Khan and Senhadji, 2001; Mubarik, 2005; Ahmed and Mortaza, 2005; Pollin and Zhu, 2005; Risso and Carrera, 2009; Singh, 2010), but more interestingly, the inflation threshold level in Rwanda falls within the ranges of the estimated thresholds in some specific country and panel studies for African countries.<sup>5</sup> For country-specific studies, the lowest threshold found was for South Africa (4%), while the highest was for Ghana (22.2%).

When comparing the level of the inflation target used for monetary policy purposes in Rwanda – around 5% – with average inflation rates and the corresponding output growth rates in the different sub-periods as shown in Table 2, it appears that the Rwandan economy performed better, in terms of economic growth, when the actual inflation rate was higher than the inflation target level.

The estimated threshold level of inflation, which is obviously higher than the policy inflation target, suggests that in the case of Rwanda there was still a wide range of higher inflation rates that were likely to be associated positively with economic growth. Therefore, it could be inferred that the adopted policy inflation target of 5% was too low,<sup>6</sup> which resulted in the implementation of unduly restrictive fiscal and monetary policies during the economic stabilization and structural adjustment programmes.

Some studies have pointed out that the adoption of such low policy inflation targets, especially in low-income countries where production capacity is not fully utilized and in which supply shocks are predominant, results in high opportunity costs in terms of foregone output growth and employment creation (Pollin and Zhu, 2005; Anwar and Islam, 2011).

## 6. Conclusion and policy implications

The main objective of this paper was to empirically examine the relationship between inflation and economic growth in Rwanda, and to investigate the existence of possible threshold effects between the two variables. To this end, a quadratic regression model using annual time series spanning 1968–2010 has been employed to estimate the threshold level of inflation. The result estimates of the quadratic regression model provided evidence supporting the hypothesis of a non-linear relationship in the inflation-growth nexus in Rwanda, and the existence of a threshold level of inflation above which inflation is detrimental to economic growth. The estimated threshold level is 12.7%. Moreover, the Granger causality test was performed to investigate the linear causation between inflation and growth in the regression model; the results of the test estimates revealed a unidirectional causality running from inflation to growth and no feedback effect from growth to inflation was detected, implying that the simultaneity bias would not seriously affect the coefficient of inflation in the regression model (Fischer, 1993; Khan and Senhadji, 2001).

This paper is the first attempt to address the threshold effect in the inflation-growth nexus in Rwanda. It is therefore crucial, as it provides a baseline study in search of the optimal inflation for growth in Rwanda. In addition, the findings of this research are consistent with the results of other recent similar empirical studies carried out on developing countries (Risso and Carrera, 2009; Singh, 2010), and in particular with results obtained in panel and specific country studies for African economies (Combey and Nubukpo, 2010; Frimpong and Oteng-Abayie, 2010; Quartey, 2010; Salami and Kelikume, 2010; Seleteng, Bittencourt and van Eyden, 2011; Leshoro, 2012; Ghazouani, 2012).

The analysis of this paper has important policy implications. First, the findings of the study revealed a significant difference between the inflation target used for policy purposes in Rwanda and the estimated inflation threshold. Since the monetary authorities have been targeting an inflation level of around 5% in the implementation of economic stabilization and structural adjustment programmes, one could infer in light of this research that monetary and fiscal policies applied in these programmes were tighter than necessary and that there was room for higher economic growth with a higher inflation rate without pushing the economy into an inflationary spiral.

It follows that, while the primary objective of monetary policy implemented by the National Bank of Rwanda is the achievement and maintenance of price stability, the monetary authorities should also be mindful of the trade-off between inflation, growth and employment in a developing country such as Rwanda in which the production capacity is not fully utilized and where supply shocks are predominant. Second, on the

basis of empirical evidence, this study suggests that the upper limit of the inflation target for policy purposes in Rwanda should be 12.7%, rather than 5%. This implies that more relaxed monetary policy should be pursued when inflation is lower than the threshold level, while a tighter stance should be adopted if inflation approaches that level.

Finally, although the objective of this research was to shed light on the relationship between inflation and growth in Rwanda, some important related issues have not been addressed, such as the channels through which inflation exerts a negative effect on economic growth and the impact of inflation variability on output growth. These are crucial issues that deserve substantial additional attention and should be investigated in future studies.

### **Notes**

- 1. Other empirical studies using cross-sectional or panel data may have included Rwanda in their sample, but unfortunately the full list of countries was not provided.
- Members of SADC are: Angola, Botswana, Democratic Republic of Congo, Lesotho, Madagascar, Malawi, Mauritius, Mozambique, Namibia, Seychelles, South Africa, Swaziland, Tanzania, Zambia and Zimbabwe.
- 3. The definition of inflation used in this paper is the headline inflation calculated on the basis of the Consumer Price Index (CPI), regularly published by the National Bank of Rwanda and the National Institute of Statistics of Rwanda. This definition was adopted for two main reasons: it is the most frequently used in empirical studies and it is available for the entire period covered by this research. The core inflation is also estimated by the National Bank of Rwanda for monetary policy purposes, however, it was first published only in 2004.
- 4. Government is defined as the ratio between Government expenditure and nominal GDP.
- 5. These studies dealt with the specific cases of Egypt (Kheir-El-Din and Abou-Ali, 2008); Ghana (Frimpong and Oteng-Abayie, 2010; Quartey, 2010); Lesotho (Seleteng, 2005); Nigeria (Salami and Kelikume, 2010) and South Africa (Leshoro, 2012). They also include panel studies: WAEMU countries (Combey and Nubukpo, 2010), SADC region (Seleteng et al., 2011) and MENA region (Ghazouani, 2012).
- 6. Even IMF economists recognized in a recent study that a wider range of inflation targets of 5–10% would be supportive of economic growth in low-income countries.

### References

- Ahmed, S. and G. Mortaza. 2005. "Inflation and economic growth in Bangladesh: 1981-2005". Working Paper: WP 0604, Research Department, Bangladesh Bank.
- Ahortor, C.R.K., A. Adenekan and W. Ohemeng. 2012. "An estimate of inflation threshold in the WAMZ: The case for Ghana and Nigeria". *Journal of Monetary and Economic Integration*, 11:158–201.
- Akerlof, G.A., W. Dickens and G. Parry. (1996). "The macroeconomics of low inflation". Brookings Papers on Economic Activity, 27: 1–26.
- Anwar, S. and I. Islam. 2011. "Should developing countries target low, single digit inflation to promote growth and employment? Employment Working Paper No. 87. International Labor Office.
- Ayisi, R. 2013. "Single digit inflation and economic growth". *American Journal of Economics* and Business Administration, 5: 22–28.
- Baer, W. 1967. "The inflation controversy in Latin America: A survey". Latin American Research Review, 2: 3–25.
- Barro, R.J. 1995. "Inflation and economic growth". Working Paper No. 5326. National Bureau of Economic Research.
- Bhatia, R.J. 1960. "Inflation, Deflation and Economic Development". IMF Staff Papers, 8: 101–114. International Monetary Fund.
- Blanchard, O.J. and N. Kiyotaki. 1987. "Monopolistic competition and the effects of aggregate demand". American Economic Review, 77: 647–666.
- Boyd, J.H., R.E. Levine and B.D. Smith. 1996. "Inflation and financial market performance". Working Paper No. 9617. Federal Reserve Bank of Cleveland.
- Bruno, M. and W. Easterly. 1998. "Inflation crises and long-run growth". *Journal of Monetary Economics*, 41: 3–26.
- Chimobi, O.P. 2010. "Inflation and economic growth in Nigeria". *Journal of Sustainable Development*, 3: 159–166.
- Choi, S., B. Smith and J.H. Boyd. 1996. "Inflation, financial markets, and capital formation". *Federal Reserve Bank of St. Louis Review*, 78: 41–58.
- Clements, B., R. Bhattacharya and T.Q Nguyen. 2005. "Can debt relief boost growth in poor countries". *IMF Economic Issues*, No. 34.
- Combey, A. and K. Nubukpo. 2010. "*Effets Non Linéaires de l'inflationsur la Croissance dans l'UEMOA*". Centre de Recherche et de Formation en Sciences Economiques et de Gestion. Université de Lomé.
- De Gregorio, J. 1992. "Effects of inflation on growth: lessons from Latin America". European Economic Review, 36: 417-425.
- De Gregorio, J. 1996. *Inflation, Growth and Central Banks: Theory and Evidence*. Ministry of Finance Paper. Santiago, Chile.
- Devarajan, S., V. Swaroop and H. Zou. 1996. "The composition of public expenditure and economic growth". *Journal of Monetary Economics*, 37: 313–344.
- Dollar, D. 1992. "Outward-oriented developing economies really do grow more rapidly: Evidence

from 95 LDCs, 1976–1985". Economic Development and Cultural Change, 40: 523–544.

- Dornbusch, R., S. Fischer and C. Kearney. 1996. *Macroeconomics*. The McGraw-Hill Companies, Inc., Sydney.
- Dorrance, G.S. 1966. "Inflation and Growth: The Statistical Evidence". IMF Staff Papers, 13: 82–102. International Monetary Fund.
- Edwards, S. 1998. "Openness, productivity and growth: what do we really know?". Economic Journal, 108: 383-398.
- Espinoza, R.H., H. Leon and A. Prasad. (2010). "Estimating the inflation-growth nexus. A smooth transition model". IMF Working Papers, WP/10/76. International Monetary Fund.
- Fama, E. and G.Schwert. 1977. "Asset returns and inflation". *Journal of Financial Economics*, 5: 115–146.
- Feldstein, M. 1982. "Inflation and tax rules and investment: Some econometric evidence". *Econometrica*, 50: 825–862.
- Fischer, S. 1993. "The role of macroeconomic factors in growth". *Journal of Monetary Economics*, 32: 485–512.
- Fischer, S., R. Sahay and C. Vegh. 1996. "Stabilization and growth in transition economies: The early experience". *Journal of Economic Perspectives*, 10: 46–66.
- Frimpong, J.M. and E.F. Oteng-Abayie. 2010. "When is inflation harmful? Estimating the threshold effect for Ghana". *American Journal of Economics and Business Administration*, 2: 225–232.
- Georgescu-Roegen, N. 1970. "Structural inflation-lock and balanced growth". *Economies et Societes*, 4: 557–605.
- Ghazouani, S. 2012. "Threshold effect of inflation on growth: evidence from MENA region". Working Paper No. 715. Economic Research Forum.
- Ghosh, A. and S. Phillips. 1998. "*Warning: Inflation may be Harmful to your Growth*". IMF Staff Papers, 45: 672–710. International Monetary Fund.
- Gillman, M., M. Harris and L. Matyas. 2004. "Inflation and growth: Explaining a negative effect". *Empirical Economics*, 29: 149–167.
- Grossman, G. and E. Helpman. 1991. *Innovation and Growth in the Global Economy*. Cambridge, MA: MIT Press.
- Hansen, B. 2000. "Sample splitting and threshold estimation". Econometrica, 68: 575-603.
- Hansen, B.1996. "Inference when nuisance parameter is not identified under the null hypothesis". *Econometrica*, 64: 413–430.
- Haslag, J. and J. Koo. 1999. "Financial repression, financial development and economic growth". Working Paper 99-02. Federal Reserve Bank of Atlanta.
- Hellerstein, R. 1997. "The impact of inflation". *Regional Review*. Federal Reserve Bank of Boston, January: 18–24.
- Hendry, D.F. 1995. Dynamic Econometrics. Oxford University Press. Oxford.
- Hermes, N. and R. Lensink. 2001. "Fiscal Policy and Private Investment in Less Developed Countries". UNU-WIDER Discussion Paper No. 2001/32.
- Hussain, M. 2005. "Inflation and growth: Estimation of threshold point for Pakistan". Economic Policy Department, State Bank of Pakistan.
- Hussain, S. and S. Malik. 2011. "Inflation and growth: Evidence from Pakistan". *International Journal of Economics and Finance*, 3: 262–276.
- Huybens, E. and B. Smith. 1998. "Financial market frictions, monetary policy, and capital accumulation in a small open economy". *Journal of Economic Theory*, 81: 353–400.
- Johanson, H.G. 1967. "Is inflation a retarding factor in economic growth?" In Krivine, D, ed., Fiscal and Monetary Problems in Developing States. Proceedings of the Third Rehoroth Conference: 121–130. New York: Praeger.
- Jones, L. and R. Manuelli. 1993. "Growth and the effects of inflation". NBER Working Paper Series No. 4523. National Bureau of Economic Research.
- Kelley, A.C. 1988. "Population pressures, savings and investment in the Third World: Some

puzzles". Economic Development and Cultural Change, 36: 449-64.

- Khan, M.S. and S.A. Senhadji. 2001. *Threshold Effects in the Relationship between Inflation and Growth*. IMF Staff Papers, 48: 1–21. International Monetary Fund, Washington.
- Khan, M.S. and S.A. Senhadji. 2000. "Threshold effects in the relationship between inflation and growth". WP/00/110, IMF Working Paper, June.
- Kheir El-Din, H. and H. Abou-Ali. 2008. "Inflation and growth in Egypt: is there a threshold effect?" ECES Working Paper No. 135.
- Kiley, M. 2000. "Endogenous price stickiness and business cycle persistence". *Journal of Money, Credit and Banking*, 32: 28–53.
- King, R. and R.Levine. 1993. "Finance and growth: Schumpeter might be right". Quarterly Journal of Economics, 108: 717-737.
- Kormendi, R.C. and P.G. Meguire. 1985. "Macroeconomic determinants of growth: Cross-country evidence". *Journal of Monetary Economics*, 16: 141–163.
- Kremer, S., A. Bick and D. Nautz. 2009. Inflation and Growth: New Evidence from a Dynamic Panel Threshold Analysis". SFB 649 Discussion Paper 2009-036. Sonderforschungsbereich, Berlin.
- Leshoro, T.L.A. 2012. "Estimating inflation threshold for South Africa". Working Paper No. 285. Economic Research South Africa.
- Levine, R.1997. "Financial development and economic growth; views and agenda". Journal of Economic Literature, 35: 688-726.
- Levine, R. and D. Renelt. 1992. "A sensitivity analysis of cross-country growth regressions". American Economic Review, 82: 942–63.
- Levine, R. and S. Zervos. 1998. "Stock markets, banks and economic growth". *American Economic Review*, 88: 537-558.
- Levine, R., N. Loayza and T. Beck. 2000. "Financial intermediation and growth: causality and causes". Journal of Monetary Economics, 46: 31-77.
- Lin, S. and H. Ye. 2009. "Does inflation targeting make a difference in developing countries?" *Journal of Development Economics*, 89: 118–123.
- Lucas, R.E. 1973. "Some international evidence on output-inflation trade-offs". *American Economic Review*, 63: 326–34.
- Mankiw, N., D. Romer and D. Weil. 1992. "Contribution to the empirics of economic growth". *The Quarterly Journal of Economics*, 2: 407–437.
- Motely, B. 1998. "Growth and inflation: A cross-country study". *Federal Reserve Bank of San Francisco Economic Review*, 1: 15–28.
- Mubarik, Y.A. 2005. "Inflation and growth: An estimate of the threshold level of inflation in Pakistan". *State Bank of Pakistan Research Bulletin*, 1: 35–44.
- Mundell, R. 1965. "Growth, stability and inflationary finance". *Journal of Political Economy*, 73: 97–109.
- Pattillo, C., Poirson and R.A. Ricci. 2002. "External debt and growth". IMF Working Paper 02/69. International Monetary Fund, Washington.
- Pollin, R. and A. Zhu. 2005. "Inflation and economic growth: A cross-country non-linear analysis". Working Paper Series No. 109. PERI, University of Massachusetts Amherst.
- Quartey, P. 2010. "Price stability and the growth maximizing rate of inflation in Ghana". *Modern Economy*, 1: 180–194.
- Risso, W.A. and E.J. Carrera. 2009. "Inflation and Mexican economic growth: Long-run relation and threshold effects". *Journal of Financial Economic Policy*, 1: 246–263.
- Romer, D. 2001. *Advanced Macroeconomics* 2<sup>nd</sup> edition. McGraw-Hill Higher Education. New York.
- Romer, P. 1990. "Endogenous technological change". Journal of Political Economy, 98: S71-S102.

- Rousseau, P. and P. Wachtel. 2002. "Inflation thresholds and the finance-growth nexus". *Journal* of International Money and Finance, 21: 777–793.
- Sachs, J.D. and A.M. Warner. 1995. "Natural resource abundance and economic growth". NBER Working Paper No. 5398.
- Sala-i-Martin, X. 1997. "I just ran two million regressions". *American Economic Review*, 87: 178–183.
- Salami, D. and I. Kelikume. 2010. "An estimation of inflation threshold for Nigeria 1970–2008". Lagos Business School.
- Sarel, M. 1996. "Non-linear Effects of Inflation on Economic Growth". IMF Staff Papers, 43: 199–215. International Monetary Fund, Washington.
- Seleteng, M., M. Bittencourt and R. van Eyden. 2011. "Non-linearities in inflation-growth nexus in the SADC Region: A panel smooth transition regression approach". Working Paper /2011/26. University of Pretoria, Pretoria.
- Seleteng, M. 2005. "Inflation and economic growth: An estimate of an optimal level of inflation in Lesotho". Research Department. Central Bank of Lesotho. Available at www.centralbank.ls
- Sepehri, A. and S. Moshiri. 2004. "Inflation-growth profiles across countries: Evidence from developing and developed countries". *International Review of Applied Economics*, 18: 191–207.
- Singh, P. 2010. "Searching threshold inflation for India". Economics Bulletin, 30: 3209-3220.
- Taylor, L. 1983. *Structuralist Macroeconomics: Applicable models for the Third World*. New York: Basic.
- Temple, J. 2000. "Inflation and growth: Stories short and tall". *Journal of Economic Surveys*, 14: 395–426.
- Thirlwall, A.P. 2000. "Trade agreements, trade liberalization and economic growth: a selective survey". African Development Review, 12: 129-160.
- Tobin, J. 1965. "Money and economic growth". Econometrica, 33: 671-684.
- Todaro, M. 1996. *Economic Development* 6<sup>th</sup> edition. Reading, Mass.: Addison Wesley Publishing Company.
- World Bank Development Indicators. 2011. World Bank.

#### **Other recent publications in the AERC Research Papers Series:**

- *Female Labour Force Participation in Ghana: The Effects of Education*, by Harry A. Sackey, Research Paper 150.
- *The Integration of Nigeria's Rural and Urban Foodstuffs Market*, by Rosemary Okoh and P.C. Egbon, Research Paper 151.
- Determinants of Technical Efficiency Differentials amongst Small- and Medium-Scale Farmers in Uganda: A Case of Tobacco Growers, by Marios Obwona, Research Paper 152.
- Land Conservation in Kenya: The Role of Property Rights, by Jane Kabubo-Mariara, Research Paper 153.
- *Technical Efficiency Differentials in Rice Production Technologies in Nigeria,* by Olorunfemi Ogundele, and Victor Okoruwa, Research Paper 154.
- *The Determinants of Health Care Demand in Uganda: The Case Study of Lira District, Northern Uganda, by Jonathan Odwee, Francis Okurut and Asaf Adebua, Research Paper 155.*
- Incidence and Determinants of Child Labour in Nigeria: Implications for Poverty Alleviation, by Benjamin C. Okpukpara and Ngozi Odurukwe, Research Paper 156.
- *Female Participation in the Labour Market: The Case of the Informal Sector in Kenya,* by Rosemary Atieno, Research Paper 157.
- *The Impact of Migrant Remittances on Household Welfare in Ghana,* by Peter Quartey, Research Paper 158.
- *Food Production in Zambia: The Impact of Selected Structural Adjustments Policies,* by Muacinga C.H. Simatele, Research Paper 159.
- Poverty, Inequality and Welfare Effects of Trade Liberalization in Côte d'Ivoire: A Computable General Equilibrium Model Analysis, by Bédia F. Aka, Research Paper 160.
- *The Distribution of Expenditure Tax Burden before and after Tax Reform: The Case of Cameroon,* by Tabi Atemnkeng Johennes, Atabongawung Joseph Nju and Afeani Azia Theresia, Research Paper 161.
- *Macroeconomic and Distributional Consequences of Energy Supply Shocks in Nigeria,* by Adeola F. Adenikinju and Niyi Falobi, Research Paper 162.
- Analysis of Factors Affecting the Technical Efficiency of Arabica Coffee Producers in Cameroon, by Amadou Nchare, Research Paper 163.
- Fiscal Policy and Poverty Alleviation: Some Policy Options for Nigeria, by Benneth O. Obi, Research Paper 164.
- FDI and Economic Growth: Evidence from Nigeria, by Adeolu B. Ayanwale, Research Paper 165.
- An Econometric Analysis of Capital Flight from Nigeria: A Portfolio Approach, by Akanni Lawanson, Research Paper 166.
- Extent and Determinants of Child Labour in Uganda, by Tom Mwebaze, Research Paper 167.
- *Oil Wealth and Economic Growth in Oil Exporting African Countries,* by Olomola Philip Akanni, Research Paper 168.
- *Implications of Rainfall Shocks for Household Income and Consumption in Uganda,* by John Bosco Asiimwe, Research Paper 169.
- *Relative Price Variability and Inflation: Evidence from the Agricultural Sector in Nigeria,* by Obasi O. Ukoha, Research Paper 170.
- A Modelling of Ghana's Inflation: 1960–2003, by Mathew Kofi Ocran, Research Paper 171.
- *The Determinants of School and Attainment in Ghana: A Gender Perspective*, by Harry A. Sackey, Research Paper 172.
- Private Returns to Education in Ghana: Implications for Investments in Schooling and Migration, by Harry A. Sackey, Research Paper 173.
- *Oil Wealth and Economic Growth in Oil Exporting African Countries,* by Olomola Philip Akanni, Research Paper 174.

- *Private Investment Behaviour and Trade Policy Practice in Nigeria*, by Dipo T. Busari and Phillip C. Omoke, Research Paper 175.
- Determinants of the Capital Structure of Ghanaian Firms, by Jochua Abor, Research Paper 176.
- *Privatization and Enterprise Performance in Nigeria: Case Study of some Privatized Enterprises,* by Afeikhena Jerome, Research Paper 177.
- Sources of Technical Efficiency among Smallholder Maize Farmers in Southern Malawi, by Ephraim W. Chirwa, Research Paper 178.
- Technical Efficiency of Farmers Growing Rice in Northern Ghana, by Seidu Al-hassan, Research Paper 179.
- *Empirical Analysis of Tariff Line-Level Trade, Tariff Revenue and Welfare Effects of Reciprocity under an Economic Partnership Agreement with the EU: Evidence from Malawi and Tanzania,* by Evious K. Zgovu and Josaphat P. Kweka, Research Paper 180.
- *Effect of Import Liberalization on Tariff Revenue in Ghana,* by William Gabriel Brafu-Insaidoo and Camara Kwasi Obeng, Research Paper 181.
- *Distribution Impact of Public Spending in Cameroon: The Case of Health Care,* by Bernadette Dia Kamgnia, Research Paper 182.
- Social Welfare and Demand for Health Care in the Urban Areas of Côte d'Ivoire, by Arsène Kouadio, Vincent Monsan and MamadouGbongue, Research Paper 183.
- Modelling the Inflation Process in Nigeria, by Olusanya E. Olubusoye and Rasheed Oyaromade, Research Paper 184.
- Determinants of Expected Poverty Among Rural Households in Nigeria, by O.A. Oni and S.A. Yusuf, Research Paper 185.
- Exchange Rate Volatility and Non-Traditional Exports Performance: Zambia, 1965–1999, by Anthony Musonda, Research Paper 186.
- Macroeconomic Fluctuations in the West African Monetary Union: A Dynamic Structural Factor Model Approach, by Romain Houssa, Research Paper 187.
- Price Reactions to Dividend Announcements on the Nigerian Stock Market, by Olatundun Janet Adelegan, Research Paper 188.
- Does Corporate Leadership Matter? Evidence from Nigeria, by Olatundun Janet Adelegan, Research Paper 189.
- Determinants of Child Labour and Schooling in the Native Cocoa Households of Côte d'Ivoire, by Guy Blaise Nkamleu, Research Paper 190.
- Poverty and the Anthropometric Status of Children: A Comparative Analysis of Rural and Urban Household in Togo, by Kodjo Abalo, Research Paper 191.
- African Economic and Monetary Union (WAEMU), by Sandrine Kablan, Research Paper 192.

*Economic Liberalization, Monetary and Money Demand in Rwanda: 1980–2005, by Musoni J. Rutayisire, Research Paper 193.* 

- Determinants of Employment in the Formal and Informal Sectors of the Urban Areas of Kenya, by Wambui R. Wamuthenya, Research Paper 194.
- An Empirical Analysis of the Determinants of Food Imports in Congo, by Léonard Nkouka Safoulanitou and Mathias Marie AdrienNdinga, Research Paper 195.
- Determinants of a Firm's Level of Exports: Evidence from Manufacturing Firms in Uganda, by Aggrey Niringiye and Richard Tuyiragize, Research Paper 196.
- Supply Response, Risk and Institutional Change in Nigerian Agriculture, by Joshua Olusegun Ajetomobi, Research Paper 197.
- Multidimensional Spatial Poverty Comparisons in Cameroon, by Aloysius Mom Njong, Research Paper 198.
- Earnings and Employment Sector Choice in Kenya, by Robert Kivuti Nyaga, Research Paper 199.
- *Convergence and Economic Integration in Africa: the Case of the Franc Zone Countries,* by Latif A.G. Dramani, Research Paper 200.

- Analysis of Health Care Utilization in Côte d'Ivoire, by Alimatou Cissé, Research Paper 201.
- *Financial Sector Liberalization and Productivity Change in Uganda's Commercial Banking Sector,* by Kenneth Alpha Egesa, Research Paper 202.
- Competition and Performance in Uganda's Banking System by Adam Mugume, Research Paper 203
- Parallel Market Exchange Premiums and Customs Revenue in Nigeria, by Olumide S. Ayodele and Francis N. Obafemi, Research Paper 204
- Fiscal Reforms and Income Inequality in Senegal and Burkina Faso: A Comparative Study, by Mbaye Diene, Research Paper 205.
- Factors Influencing Technical Efficiencies among Selected Wheat Farmers in UasinGishu District, Kenya, by James Njeru, Research Paper 206.
- *Exact Configuration of Poverty, Inequality and Polarization Trends in the Distribution of wellbeing in Cameroon,* by Francis Menjo Baye, Research Paper 207.
- *Child Labour and Poverty Linkages: A Micro Analysis from Rural Malawian Data,* by Leviston S. Chiwaula, Research Paper 208.
- The Determinants of Private Investment in Benin: A Panel Data Analysis, by Sosthène Ulrich Gnansounou, Research Paper 209.
- Contingent Valuation in Community-Based Project Planning: The Case of Lake Bamendjim Fishery Re-Stocking in Cameroon, by William M. Fonta, Hyacinth E. Ichoku and Emmanuel Nwosu, Research Paper 210.
- *Multidimensional Poverty in Cameroon: Determinants and Spatial Distribution*, by Paul Ningaye, Laurent Ndjanyou and Guy Marcel Saakou, Research Paper 211.
- What Drives Private Saving in Nigeria, by Tochukwu E. Nwachukwu and Peter Odigie, Research Paper 212.
- *Board Independence and Firm Financial Performance: Evidence from Nigeria*, by Ahmadu U. Sanda, Tukur Garba and Aminu S. Mikailu, Research Paper 213.
- *Quality and Demand for Health Care in Rural Uganda: Evidence from 2002/03 Household Survey,* by Darlison Kaija and Paul Okiira Okwi, Research Paper 214.
- Capital Flight and its Determinants in the Franc Zone, by Ameth Saloum Ndiaye, Research Paper 215.
- *The Efficacy of Foreign Exchange Market Intervention in Malawi*, by Kisukyabo Simwaka and Leslie Mkandawire, Research Paper 216.
- The Determinants of Child Schooling in Nigeria, by Olanrewaju Olaniyan, Research Paper 217. Influence of the Fiscal System on Income Distribution in Regions and Small Areas: Microsimulated
- CGE Model for Côte d'Ivoire, by Bédia F. Aka and Souleymane S. Diallo, Research Paper 218.
- Asset Price Developments in an Emerging Stock Market: The Case Study of Mauritius by Sunil K. Bundoo, Research Paper 219.
- IntrahouseholdResources Allocation in Kenya by Miriam Omolo, Research Paper 220.
- Volatility of Resources Inflows and Domestic Investment in Cameroon by Sunday A. Khan, Research Paper 221.
- *Efficiency Wage, Rent-Sharing Theories and Wage Determination in Manufacturing Sector in Nigeria* by Ben E. Aigbokhan, Research Paper 222.
- *Government Wage Review Policy and Public-Private Sector Wage Differential in Nigeria* by Alarudeen Aminu, Research Paper 223.
- Rural Non-Farm Incomes and Poverty Reduction in Nigeria by Awoyemi Taiwo Timothy, Research Paper 224.
- *After Fifteen Year Use of the Human Development Index (HDI) of the United Nations Development Program (UNDP): What Shall We Know?* by Jean Claude Saha, Research Paper 225.
- Uncertainty and Investment Behavior in the Democratic Republic of Congo by Xavier Bitemo Ndiwulu and Jean-Papy Manika Manzongani, Research Paper 226.

- An Analysis of Stock Market Anomalies and Momentum Strategies on the Stock Exchange of Mauritius by Sunil K. Bundoo, Research Paper 227.
- *The Effect of Price Stability on Real Sector Performance in Ghana* by Peter Quartey, Research Paper 228.
- The Impact of Property Land Rights on the Production of Paddy Rice in the Tillabéry, Niamey and Dosso Regions in Niger by Maman Nafiou Malam Maman and Boubacar Soumana, Research Paper 229.
- An Econometric Analysis of the Monetary Policy Reaction Function in Nigeria by Chukwuma Agu, Research Paper 230.
- Investment in Technology and Export Potential of Firms in Southwest Nigeria by John Olatunji Adeoti, Research Paper 231.
- Analysis of Technical Efficiency Differentials among Maize Farmers in Nigeria by Luke OyesolaOlarinde, Research Paper 232.
- Import Demand in Ghana: Structure, Behaviour and Stability by Simon Kwadzogah Harvey and Kordzo Sedegah, Research Paper 233.
- Trade Liberalization Financing and Its Impact on Poverty and Income Distribution in Ghana by Vijay K. Bhasin, Research Paper 234.
- An Empirical Evaluation of Trade Potential in Southern African Development Community by Kisukyabo Simwaka, Research Paper 235.
- Government Capital Spending and Financing and Its Impact on Private Investment in Kenya: 1964-2006 by Samuel O. Oyieke, Research Paper 236.
- Determinants of Venture Capital in Africa: Cross Section Evidence by Jonathan Adongo, Research Paper 237.
- Social Capital and Household Welfare in Cameroon: A Multidimensional Analysis by Tabi Atemnkeng Johannes, Research Paper 238.
- Analysis of the Determinants of Foreign Direct Investment Flows to the West African and Economic Union Countries by Yélé Maweki Batana, Research Paper 239.
- Urban Youth Labour Supply and the Employment Policy in Côte d'Ivoire by Clément Kouadio Kouakou, Research Paper 240.
- Managerial Characteristics, Corporate Governance and Corporate Performance: The Case of Nigerian Quoted Companies by Adenikinju Olayinka, Research Paper 241.
- *Effects of Deforestation on Household Time Allocation among Rural Agricultural Activities: Evidence from Western Uganda* by Paul Okiira Okwi and Tony Muhumuza, Research Paper 242.
- The Determinants of Inflation in Sudan by Kabbashi M. Suliman, Research Paper 243.
- Monetary Policy Rules: Lessons Learned From ECOWAS Countries by Alain Siri, Research Paper 244.
- Zimbabwe's Experience with Trade Liberalization by Makochekanwa Albert, Hurungo T. James and Kambarami Prosper, Research Paper 245.
- Determinants in the Composition of Investment in Equipment and Structures in Uganda by Charles Augustine Abuka, Research Paper 246.
- *Corruption at Household Level in Cameroon: Assessing Major Determinants* by Joseph-Pierre Timnou and Dorine K. Feunou, Research Paper 247.
- *Growth, Income Distribution and Poverty: The Experience of Côte d'Ivoire from 1985 to 2002* by Kouadio Koffi Eric, Mamadou Gbongue and Ouattara Yaya, Research Paper 248.
- Does Bank Lending Channel Exist In Kenya? Bank Level Panel Data Analysis by Moses Muse Sichei and Githinji Njenga, Research Paper 249.
- *Governance and Economic Growth in Cameroon* by Fondo Sikod and John NdeTeke, Research Paper 250.
- *Analyzing Multidimensional Poverty in Guinea: A Fuzzy Set Approach* by Fatoumata Lamarana Diallo, Research Paper 251.

- The Effects of Monetary Policy on Prices in Malawi by Ronald Mangani, Research Paper 252.
- Total Factor Productivity of Agricultural Commodities in the Economic Community of West African States: 1961-2005 by Joshua Olusegun Ajetomobi, Research Paper 253.
- Public Spending and Poverty Reduction in Nigeria: A Benefit Incidence Analysis in Education and Health by Uzochukwu Amakom, Research Paper 254.
- Supply Response of Rice Producers in Cameroon: Some Implications of Agricultural Trade on Rice Sub-sector Dynamics by Ernest L. Moluaand Regina L. Ekonde, Research Paper 255.
- *Effects of Trade Liberalization and Exchange Rate Changes on Prices of Carbohydrate Staples in Nigeria* by A. I. Achike, M. Mkpado and C. J. Arene, Research Paper 256.
- *Underpricing of Initial Public Offerings on African Stock Markets: Ghana and Nigeria* by Kofi A. Osei, Charles K.D. Adjasiand Eme U. Fiawoyife, Research Paper 257.
- Trade Policies and Poverty in Uganda: A Computable General Equilibrium Micro Simulation Analysis by Milton Ayoki, Research Paper 258.
- Interest Rate Pass-through and Monetary Policy Regimes in South Africa by Meshach Jesse Aziakpono and Magdalene Kasyoka Wilson, Research Paper 259.
- *Vertical Integration and Farm gate prices in the Coffee Industry in Côte d'Ivoire* by Malan B. Benoit, Research Paper 260.
- Patterns and Trends of Spatial Income Inequality and Income Polarization in Cameroon by Aloysius Mom Njong and Rosy Pascale MeyetTchouapi, Research Paper 261.
- Private Sector Participation in the Provision of Quality Drinking Water in Urban Areas in Ghana: Are the People Willing to Pay? By Francis Mensah Asenso-Boadi and Godwin K. Vondolia, Research Paper 262.
- Private Sector Incentives and Bank Risk Taking: A Test of Market Discipline Hypothesis in Deposit Money Banks in Nigeria by Ezema Charles Chibundu, Research Paper 263.
- A Comparative Analysis of the Determinants of Seeking Prenatal Health Care in Urban and Rural Areas of Togo by Ablamba Johnson, AlimaIssifouand EtsriHomevoh, Research Paper 264.
- Predicting the Risk of Bank Deterioration: A Case Study of the Economic and Monetary Community of Central Africa by Barthélemy Kouezo, Mesmin Koulet-Vickot and Benjamin Yamb, Research Paper 265.
- Analysis of LabourMarket Participation in Senegal by Abou Kane, Research Paper 266.
- What Influences Banks' Lending in Sub-Saharan Africa? by Mohammed Amidu, Research Paper 267.
- *Central Bank Intervention and Exchange Rate Volatility in Zambia* by Jonathan Mpundu Chipili, Research Paper 268.
- *Capital Flight from the Franc Zone: Exploring the Impact on Economic Growth* by Ameth Saloum Ndiaye, Research Paper 269.
- Dropping Out of School in the Course of the Year in Benin: A Micro-econometric Analysis by Barthelemy M. Senou, Research Paper 270.
- Determinants of Private Investment Behaviour in Ugandan Manufacturing Firms by Niringyiye Aggrey, Research Paper 271.
- Determinants of Access to Education in Cameroon by Luc Nembot Ndeffo, Tagne Kuelah Jean Réné & Makoudem Téné Marienne Research Paper 272.
- *Current Account Sustainability in the West African Economic and Monetary Union Countries* by Zakarya Keita, Research Paper 273.
- An Empirical Assessment of Old Age Support in sub-Saharan Africa: Evidence from Ghana by Wumi Olayiwola, Olusanjo Oyinyole& S.L. Akinrola, Research Paper 274.
- Characteristics and Determinants of Child Labour in Cameroon by Laurent Ndjanyou and Sébastien Djiénouassi, Research Paper 275.
- *Private Sector Investment in Sierra Leone: An Analysis of the Macroeconomic Determinants* by Mohamed Jalloh, Research Paper 276.

- *Technical Efficiency and Productivity of Primary Schools in Uganda* by Bruno L. Yawe, Research Paper 277.
- Comparisons of Urban and Rural Poverty Determinants in Cameroon by Samuel Fambon, Research Paper 278.
- Impact of External Debt Accumulation and Capital Flight on Economic Growth of West African Countries, by Akanni O. Lawanson, Research Paper 279.
- *Female Education and Maternal Health Care Utilization in Uganda,* by Asumani Guloba and Edward Bbaale, Research Paper 280.
- *HIV/AIDS Sero-prevalence and Socio-economic Status: Evidence from Uganda,* by Ibrahim Kasirye, Research Paper 281.
- *Female Education, Labour Force Participation and Fertility: Evidence from Uganda,* by Edward Bbaale, Research Paper 282.
- *Estimating the Informal Cross-border Trade in Central Africa,* by Robert Nkendah, Chantal Beatrice Nzouessin and Njoupouognigni Moussa, Research Paper 283.
- Health and Economic Growth in Sub-Saharan African Countries, by Eric Kehinde Ogunleye, Research Paper 284.
- *Effects of Collective Marketing by Farmers' Organizations on Cocoa Farmers' Price in Cameroon,* by Kamden Cyrille Bergaly and Melachio Tameko Andre, Research Paper 285.
- Modelling Trade Flows between Three Southern and Eastern African Regional Trade Agreements: A Case Study, by Sannassee R. Vinesh, Seetanah Boopen and Tandrayen Verena, Research Paper 286.
- Households' Incomes and Poverty Dynamics in Rural Kenya: A Panel Data Analysis, by Milu Muyanga and Phillip Musyoka, Research Paper 287.
- *The Impact of Economic Partnership Agreements between ECOWAS and the EU on Niger,* by Amadou Ousmane, Research Paper 288.
- *A Re-examination of the Determinants of Child Labour in Cote d'Ivoire,* by Edouard Pokou Abou, Research Paper 289.
- The Inflationary Effects of Fiscal Deficit in Sierra Leone: A Simulation Approach, by Robert Dauda Korsu, Research Paper 290.
- Fiscal Decentralization and Social Services in Nigeria, by Eme A. Dada, Research Paper 291.
- An Analysis of Married Women's Empowerment in Sub-Saharan Africa, by Yélé Maweki Batana and Pitaloumani Gnakou Ali, Research Paper 292.

**THIS RESEARCH STUDY** was supported by a grant from the African Economic Research Consortium. The findings, opinions and recommendations are those of the author, however, and do not necessarily reflect the views of the Consortium, its individual members or the AERC Secretariat.

Published by: The African Economic Research Consortium P.O. Box 62882 – City Square Nairobi 00200, Kenya

Printed by: The Regal Press (K) Ltd 14 Bunyala Road, Industrial Area Tel: 020 2664891 Email: info@regalpress.org Nairobi, Kenya

ISBN 978-9966-023-80-3

© 2015, African Economic Research Consortium.

## Contents

List List Abst Ackr	of tables of figures ract nowledgements	
1.	Introduction	1
2.	Literature review	4
3.	Inflation and growth trends in Rwanda	9
4.	Methodological framework	12
5.	Data, empirical results and discussion	16
6.	Conclusion and policy implications	21
Note	S	23
Refe	rences	24

## List of tables

1.	Economic indicator trends	11
2.	Inflation ranges and economic growth	11
3.	Unit root tests	17
4.	Causality tests	17
5.	Results of the quadratic regression model	19

# List of figures

1.	Inflation trends and real GDP growth	
----	--------------------------------------	--

### Abstract

It is now widely agreed among economists, policy makers and central bankers that the main objective of macroeconomic policy is to achieve a high and sustained economic growth rate while maintaining a low inflation rate. It is also believed that high inflation has an adverse effect on economic growth. But how low should the inflation rate be not to impact negatively on economic growth?

Monetary authorities in Rwanda have been targeting an inflation level of around 5% for economic policy purposes. Was this inflation target the most appropriate for economic growth? Recent studies have demonstrated that, depending on the structure and level of development of the economy, inflation becomes detrimental to economic growth when it exceeds a certain threshold. Below this threshold the impact of inflation on growth is non-significant or even positive.

Against this backdrop, this paper assumes a non-linear relationship between inflation and economic growth and attempts to identify the existence of threshold effects between these variables in the case of Rwanda using a data set spanning the sample period 1968–2010. The existence of a threshold level above which inflation has an adverse effect on economic growth in Rwanda has been investigated by means of a quadratic regression model and the ordinary least squares technique. The results showed that at low levels inflation does not hurt economic growth, while at higher levels inflation reduces economic growth. The estimated inflation threshold level is 12.7%.

As the findings of this study have unveiled the estimated threshold inflation rate consistent with economic growth in Rwanda, they would provide some useful guidance to economic decision makers in designing a more appropriate macroeconomic policy framework.

JEL: *E31; C13; 040* Key words: *Inflation; Economic growth; Threshold effects* 

# Acknowledgements

I am grateful to the African Economic Research Consortium (AERC) for providing financial support and a conducive environment to undertake and complete this study. I am heavily indebted to the resource persons and colleagues of group B for their valuable comments on previous versions of this research during the Biannual Workshops organized by the AERC. I would finally like to express my gratitude to external reviewers for their useful comments and suggestions. I nevertheless remain solely responsible for any errors in the paper.